Author's response to reviews

Title: Tetanus developed after replantation of an amputated finger: a case report

Authors:

Kenji Hayashida (tokimayu122710@gmail.com)
Chikako Murakami (cmurakami@ncmc.hosp.go.jp)
Masaki Fujioka (mfujioka@ncmc.hosp.go.jp)

Version: 8 Date: 20 August 2012

Author's response to reviews: see over
Bio Med Central Editorial team

Object: MS: 1086224059724553 - Tetanus developed after replantation of an amputated finger: a case report. Dr Kenji Hayashida et al.

Thank you for consideration of our manuscript for publication in your journal.
We have reviewed and the above manuscript according to your editor’s comments.
My manuscript has been checked over by a native English speaker and the language has been revised.